Author’s response to reviews

Title: Development of an Online Public Health Curriculum for Medical Students: The Public Health Commute

Authors:

Sarah Godfrey (sg2902@columbia.edu)
Katherine Nickerson (kgn1@cumc.columbia.edu)
Jonathan Amiel (jma2106@cumc.columbia.edu)
Benjamin Lebwohl (bl114@cumc.columbia.edu)

Version: 2 Date: 24 Jul 2019

Author’s response to reviews:

Dear Ms. Fagelson,

Thank you for consideration of our manuscript. The final document has been edited to include the appropriate titles and remove the highlighted changes. The authors initials have been included in the contributions section.

Sincerely,

Sarah Godfrey